Author's response to reviews

Title: Genomic profile predicts the efficacy of neoadjuvant chemotherapy for cervical cancer patients

Authors:

Naoki Horikawa (gorizou@kuhp.kyoto-u.ac.jp)
Tsukasa Baba (babatsu@kuhp.kyoto-u.ac.jp)
Noriomi Matsumura (noriomi@kuhp.kyoto-u.ac.jp)
Ryusuke Murakami (ryusukem@kuhp.kyoto-u.ac.jp)
Kaoru Abiko (kaoruvc@kuhp.kyoto-u.ac.jp)
Junzo Hamanishi (inkhmns@kuhp.kyoto-u.ac.jp)
Ken Yamaguchi (soulken@kuhp.kyoto-u.ac.jp)
Masafumi Koshiyama (koshiya@kuhp.kyoto-u.ac.jp)
Yumiko Yoshioka (yyumivn@kuhp.kyoto-u.ac.jp)
Ikuo Konishi (konishi@kuhp.kyoto-u.ac.jp)

Version: 4
Date: 26 August 2015

Author's response to reviews: see over
August 26, 2015

Professor Dafne Solera,
Executive Editor
BMC Cancer

Dear Professor Dafne Solera,

Thank you for editing our manuscript. We are very pleased with your interest in our paper. In accordance with the “Requesting deposition of data”, we have added the information of our deposited microarray data (GSE70035) not only in the supplementary materials but in the main text (Page 6 Line 147-8).

We believe that the revised manuscript is substantially improved and hope that you find it suitable for publication in *BMC Cancer*.

Sincerely yours,

Tsukasa Baba, M.D., Ph.D.
Department of Gynecology & Obstetrics
Kyoto University Graduate School of Medicine
54 Shogoin-Kawahara-cho, Sakyo-ku, Kyoto 606-8507, Japan
Tel. 81-75-751-3269 Fax. 81-75-761-3967
E-mail babatsu@kuhp.kyoto-u.ac.jp